Author's response to reviews

Title: Simple New Technique for Removal of Fractured Femoral Stems: a case report

Authors:

Goran Bicanic (gbic@mef.hr)
Kresimir Crnogaca (goran.bicanic@zg.t-com.hr)
Domagoj Delimar (domagoj.del@gmail.com)

Version: 2 Date: 14 February 2014

Author's response to reviews:

Dear Editor,

according to your suggestions following corrections were performed.
1. all the co-authors' email addresses included in the Title page.
2. included the age, sex, and ethnicity of the patient in the Abstract section.
3. changed the 'Case Report' section header to 'Case Presentation'.
4. included the sex of the patient in the case presentation section

Best regards,
Kresimir Crnogaca